Author's response to reviews

Title: Conservative treatment of idiopathic scoliosis according to FITS concept: presentation of the method and preliminary radiological and clinical results respecting SOSORT and SRS criteria.

Authors:

Marianna PT, PhD Bialek MB (mbfits@o2.pl)

Version: 2 Date: 8 August 2011

Author's response to reviews:

The manuscript contains the amendments proposed by the reviewers, were added two tables, extended the discussion, has been added references and changed the conclusions.